Author’s response to reviews

Title: A case report of Tinea capitis in infant in first year of life

Authors:

Narcisa Mandras (narcisa.mandras@unito.it)
Janira Roana (janira.roana@unito.it)
Ornella Cervetti (ornella.cervetti@unito.it)
Michele Panzone (mpanzone@cittadellasalute.to.it)
Vivian Tullio (vivian.tullio@unito.it)

Version: 3 Date: 04 Feb 2019

Author’s response to reviews:

Dear Editor,

we incorporate the changes suggested.

We resubmit a final modified version of our manuscript BPED-D-18_00925R2 entitled “A case report of Tinea capitis in infant in first year of life” for publication.

Changes made are highlighted in yellow in manuscript.

Best regards,

Prof. Vivian Tullio

4th February 2019

Editor Comments:
Case Presentation

Line 32: ‘However, 15 days later at the end of treatment, he presented…’
Changed as requested.

Line 35: ‘In infants, sometimes tinea capitis is misdiagnosed and underreported because it is similar…’
Done.

Line 37: ‘…from a mycological point of view to inform the differential diagnosis’ (or similar)
Done.

Background

Line 54: ‘…in children 3 to 7 years old…’
Changed as requested.

Line 61: ‘He was suspected to have a dermatophytosis.’
Changed as requested.

Case presentation

Line 76: ‘…15 days later at the end of treatment, in the…’
Changed as requested.

Discussion

Line 83: ‘…remains the most common organism responsible for TC…’
Done.

Line 86: ‘…is the most common in Greece and Belgium.’
Done.

Line 91: What is meant by ‘alopecia clinic’?
Please see the revised version page 4 line 91: “alopecia clinic” was changed in “alopecia”.

Line 101: ‘Oral terbinafine, which is often administered in dermatophytosis, is FDA approved…’

Done.

Line 104: ‘…to be as prudent an alternative due to…’

Changed as requested.